Author’s response to reviews

Title: Sodium vanadate combined with L-ascorbic acid delays disease progression, enhances motor performance, and ameliorates muscle atrophy and weakness in mice with spinal muscular atrophy

Authors:

- Huei-Chun Liu (wombatliu@gmail.com)
- Chen-Hung Ting (koichiting@gmail.com)
- Hsin-Lan Wen (landy01231977@gmail.com)
- Li-Kai Tsai (milikai@ntuh.gov.tw)
- Hsiu-Mei Hsieh-Li (hmhsieh@ntnu.edu.tw)
- Hung Li (hungli@gate.sinica.edu.tw)
- Sue Lin-Chao (mbsue@gate.sinica.edu.tw)

Version: 7 Date: 28 December 2012

Author’s response to reviews: see over
Dear Dr. Lee,

Enclosed please find our revised manuscript (MS: 9610239758122127). In response to your concerns, we have removed all issue number from the reference list, and have included all authors’ names for each citation if there are fewer than 30 authors. In addition, Table 1 and 2 were created by using a Tables tool for the revision.

Thank you very much for your correction and we hope that you will find the manuscript suitable for publication in *BMC Medicine*. We look forward to hearing from you.

Sincerely,

Sue Lin-Chao, Ph.D.
Distinguished Research Fellow
Institute of Molecular Biology, Academia Sinica
Taipei, Taiwan